Author's response to reviews

Title: Participant recruitment and retention in a pilot program to prevent weight gain in low-income overweight and obese mothers

Authors:

Mei-Wei Chang (changme@msu.edu)
Roger Brown (rlbrown3@wiscmail.wisc.edu)
Susan Nitzke (nitzke@nutrisci.wisc.edu)

Version: 6 Date: 16 November 2009

Author's response to reviews:

November 15, 2009

Dear Dr. Norton:

We are submitting the final draft of manuscript 1355829216250298 entitled: “Participant recruitment and retention in a pilot program to prevent weight gain in low-income overweight and obese mothers” for publication. We have proofread the manuscript and made required modifications.

Thanks for publishing this manuscript.

Authors: Mei-Wei Chang, PhD, RN; Roger Brown, PhD; Susan Nitzke, PhD, RD

Sincerely,

Mei-Wei Chang, PhD, RN (Corresponding author)

Michigan State University
College of Nursing
B515F West Fee Hall
East Lansing, MI 48824

e-mail: changme@msu.edu

Phone: 517-353-8682
Fax: 517-353-8536